Author's response to reviews

Title: Supervised physical exercise to improve the quality of life of cancer patients: the EFICANCER randomised controlled trial

Authors:

Aintzane Sancho (AINTZANE.SANCHO@osakidetza.net)
Sergio Carrera (SERGIO.CARRERA@osakidetza.net)
Marisol Arietaleanizbeasoa (MARIASOLEDAD.ARIETALEANIZBEASCOASARABIA@osakidetza.net)
Veronica Arce (veronica.arce@osakidetza.net)
Nere Mendizabal (NERE.MANDIZABALGALLASTEGUI@osakidetza.net)
Anna Gine (ANNA.GINEMARCH@osakidetza.net)
Aitor Sanz-Guinea (Aitor.SanzGuinea@osakidetza.net)
Araceli Eskisabel (MARIARACELI.EKISABELGONI@osakidetza.net)
Ana Lopez (ANAMARIA.LOPEZ@osakidetza.net)
Rosa A martin (ROSAAMAIA.MARTINSANTIDRIAN@osakidetza.net)
Guillermo Lopez-Vivanco (GUILLERMO.LOPEZVIVANCO@osakidetza.net)
Gonzalo Grandes (GONZALO.GRANDES@osakidetza.net)

Version: 4 Date: 7 October 2014

Author's response to reviews: see over
Dear Editorial board,

Please find attached the requested new version of the manuscript, revised according to your comments, clarifying the ethical approval obtained for our study.

(a) Ethical approval

The Ethics Committee of the Basque Country assesses all multicentre clinical trials performing at health centres from the Basque Country. Therefore, as our study is a multicentre clinical trial conducted in four health centres in the Basque Country, it is enough to have the approval by the Ethics Committee of the Basque Country.

We very much appreciate your attention to our submission

Sincerely,

Aintzane Sancho
Sergio Carrera
Marisol Arietaleainizbeascool
Verónica Arce
Nere Mendizabal Gallastegui
Anna Giné March
Aitor Sanz-Guinea
Araceli Eskisabel
Ana Lopez Rodriguez
Rosa Amaia Martin
Guillermo Lopez- Vivanco
Gonzalo Grandes